Author’s response to reviews

Title: Monophasic Synovial Sarcoma presenting as a primary ileal mass: Case Report and Literature Review.

Authors:

Alaa N Alsharif (ahmadtafigue@yahoo.com)
Musa M Fageeh (yuousofaaa@aol.com)
Yousof A Alabdulkarim (Yousof.abdulkarim@gmail.com)

Version: 2 Date: 10 August 2011

Author’s response to reviews:

based on the important comments provided by the distinguished reviewers, we made the following changes:

changed the title
re-viewed the abstract
reviewed the case presentation and provided the actual follow up period
corrected some grammatical mistakes
all changes were made with a green color to facilitate the recognition
the font was changed according to the recommended